rev 02 Sep 2003

Disclaimer for Material Taught in this Course John R. Black CSCI 6268/TLEN 5831, Network Security, Fall 2004

I understand that in this class we may cover methods to exploit vulnerabilities in contemporary computer systems and computer networks. I further understand that we may learn techniques employed by unethical individuals to circumvent security mechanisms, violate copyright, cause damage, cause financial loss, or break the law in other ways.

I hereby pledge to use all information obtained in this class in an ethical and responsible manner, properly observing University Policy and the law.

Name

Signature

Note: If you are unfamiliar with University Policy on use of its computers and networks, you should read the statement at http://www.colorado.edu/policies/cnr/. This URL is also available from the course website.